Author’s response to reviews

Title: A protocol for co-creating research project lay summaries with stakeholders: Guideline development for Canada's AGE-WELL Network

Authors:

Mineko Wada (mineko_wada@sfu.ca)
Judith Sixsmith (j.sixsmith@dundee.ac.uk)
Gail Harwood (gharwood@shaw.ca)
Theodore Cosco (theodore_cosco@sfu.ca)
Mei Fang (m.l.fang@dundee.ac.uk)
Andrew Sixsmith (andrew_sixsmith@sfu.ca)

Version: 2 Date: 09 Apr 2020

Author’s response to reviews:

RIAE-D-19-00071R1

A protocol for co-creating research project lay summaries with stakeholders: Guideline development for Canada's AGE-WELL Network

Thank you for the review of our manuscript. We list the feedback and response below.

From Co-Editors-in-Chief

“[W]e ask you to focus the title a little more on the paper as it now stands, perhaps by adding ‘in Canada's AGE-WELL project’ or similar words. Although your protocol may indeed be applicable to other situations, your paper emphasises that it is a protocol for this particular project, and we would urge that the paper's title should reflect this.”

We have changed the title to “A protocol for co-creating research project lay summaries with stakeholders: Guideline development for Canada's AGE-WELL Network.”

We hope this satisfactorily addresses your concern. Thank you again for reviewing this manuscript. Your consideration is much appreciated.